Author's response to reviews

Title: Why some physicians in Portuguese-speaking African countries work exclusively for the private sector? Findings from a mixed-methods study

Authors:

- Giuliano Russo (grusso@ihmt.unl.pt)
- Bruno de Sousa (bruno.desousa@fpce.uc.pt)
- Mohsin Sidat (mmsidat@gmail.com)
- Paulo Ferrinho (pferrinho@ihmt.unl.pt)
- Gilles Dussault (gillesdussault@ihmt.unl.pt)

Version: 2 Date: 9 June 2014

Author's response to reviews:

Please see Response to reviewers document uploaded above.